Author's response to reviews

Title: In-hospital cerebrovascular complications following orthotopic liver transplantation: A retrospective study

Authors:

Li Ling (linglirabbit@163.com)
Xiaoshun He (gdtrc@163.com)
Jinsheng Zeng (zengjs@pub.guangzhou.gd.cn)
Zhijian Liang (lzj2004152000@yahoo.com.cn)

Version: 8 Date: 1 December 2008

Author's response to reviews:

2008-12-01

Dear Dr. Sabina Alam,

Thank you very much for accepting our manuscript entitled "In-hospital cerebrovascular complications following orthotopic liver transplantation: A retrospective study" (Manuscript ID 5770425052138950). We have now revised the manuscript following the editor’s suggestions and the changes are listed below at a point-by-point style.

With best regards,

Sincerely yours,

Jinsheng Zeng, MD, PhD
Department of Neurology and Stroke Center,
First Affiliated Hospital, Sun Yat-Sen University,
No. 58 Zhongshan Road 2, Guangzhou, 510080, P.R. China.

Responses to the editor:

Formatting changes requested
----------------------------------

Major revisions (we require the author to make these changes)

Additional files - All additional files must be mentioned in the text, or removed from the system.
Response: We removed the additional files.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Remove the words 'title page' from title page.
Response: We removed the words 'title page' from title page

Remove running title.
Response: We removed the running title.

Qualifications - Please remove the authors’ qualifications (e.g. PhD / MD) from the manuscript and submissions system
Response: We removed the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system

Make sure that the author's are clearly linked to their affiliations via superscript numbers, symbols or letters.
Response: Yes, we do.

Email address - Please use the following format -
GSF: g.s.feder@qmul.ac.uk
HS: hsohal@yahoo.co.uk
SE: s.eldridge@qmul.ac.uk
Response: Thanks, we did.

Remove title and keywords from Abstract section.
Response: We removed the title and keywords from Abstract section.